Author's response to reviews

Title: On the didactics of pharmacological communication: A study of final semester health science students

Authors:

Ann Zetterqvist (ann.zetterqvist@ped.gu.se)
Patrik Aronsson (patrik.aronsson@pharm.gu.se)
Staffan Hägg (staffan.hagg@liu.se)
Karin Kjellgren (karin.kjellgren@liu.se)
Margareta Reis (margareta.reis@liu.se)
Gunnar Tobin (gunnar.tobin@pharm.gu.se)
Shirley Booth (shirley.booth@gu.se)

Version: 2 Date: 1 June 2015

Author's response to reviews:

Dear reviewers,

I have now done the following changes due to comments from the editors:

1) Inserted a short text about our ethical considerations.
2) Restructured the abstract
3) Inserted lines in the document.

Best wishes!

Ann Zetterqvist